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Highlights:

   The beamline in the wiggler region was completely hooked up this week and brought under vacuum in preparation for running the system again with our new short wavelength (1-3 micron)

electromagnetic wiggler.
   The injector was turned on and re-setup of the system begun.  We are using the opportunity to train a larger group of scientists in the machine procedures so that future measurements of the beam can be performed more expeditiously.

   The high reflector 2.8 micron mirror was mounted in the system and preparations were made for installation of the outcoupler on Monday.

   Substantial progress was made in getting the Optical Transport System (Lite) assembled and installed to bring laser beam upstairs for the first time with this laser.

   We released bid packages for the UV chicane dipoles and the sextupoles.

   We presented three talks at the DEPS conference covering the IR Upgrade performance, the High Current Injector, and an airborne FEL concept.

   Markups to the Injector Test Facility increasing the facility size and capability to maximize room accommodation were turned over to Plant Engineering for detailing. 

   We offer our congratulations to Pavel Evtushenko of Rossendorf for passing his Ph.D. oral exam and look forward to when he will join JLab in the near future. 

Management:

  We had 7 people attend the DEPS conference listening to plans and progress in Directed Energy as well as presenting our own work.
   We also had Fred Dylla, Claus Rode, and George Neil attend the MW panel Workshop in DC.  Rode spoke on cryogenic systems requirements and Neil spoke on energy recovery requirements for high power systems.

   Further work on our MOU with ONR was opportunistically done at the DEPS conference with our ONR Program Manager.  We are near to finalizing the FY'05-07 plan, tasks and budgets.  We also used the event to present our Injector Test Stand details to our review panel. 
WBS 4 (Injector):

   The gun is operational and ready for beam ops. Removed coated test tube and ground ring from high voltage test stand.  The ground ring is being re-polished to repair the damage that occurred when the test stand power supply failed during the last test in September. A new test tube was fabricated and is in the early stages of polishing.

  Have completed a list of items needed to finish the second gun assembly.

    The 7cell CEBAF cavity has been integrated successfully into our FEL PARMELA. The next step is to build an entire module (8 cavities) with this new 7cell routine and test it in PARMELA.

    Progress continues in developing a program to generate arbitrary distributions at the entrance of the linac to study the longitudinal emittance growth through the linac.

WBS 8 (Instrumentation):

   Most of the loose ends have been tied up this week (at least if feels that way). Beam Position Monitors (BPMs) in the LINAC were tested in preparation for re-commissioning that portion of the machine to run to the straight ahead dump.  Operation of the Dump Current Monitor and the Dump Water Control System was also verified. With beam operations beginning again today the punch list begins.
    We worked through our 2nd draft of a complete Machine Protection System (MPS) certification. We completed the certification for running beam to the strait-a-head (2G) dump and checked most all of the other individual pieces. Our goal for the next few days is only to run the machine in the straight ahead configuration for the purpose of doing injector setup and studies. The final certification, namely machine mode 3 (or Energy recovery mode/1G dump), will be done next week when the vacuum recovers for the shut down work. This will be done in accordance with the certification procedure that we are writing and revising.
    Our website (http://laser.jlab.org) has been updated once again and now features password protection for some of the resources. The JLab CUE login and password should be used. The computer center notified us of a potential security threat in the web video servers we use for remote viewing (Axis 2400+ at http://129.57.20.46). The video servers were taken off-line immediately. An update to the device firmware was made available by the manufacturer to correct the issues and the devices have been upgraded and are back online.
    A number of software features have been added as well as removed (bugs). Added the Photocathode Gun Coulomb Archiver to the ops menu (log was made on the update and its location). The Archiver passed its first beam test last Thursday. The results were approved by the Gun group. Started calibration process for the Beam viewers for the WesCam measurements.
    Fabrication of the four new Optical Heater Control Chassis is underway. Seven new 32:32 Channel Video Crosspoint Switcher Modules will be ready early next week to continue the upgrade of that system. The Beam viewer system is nearing completion with most viewers already ready for Ops. Alignment of the Wiggler viewers continues and will be completed Monday. The paddle viewers and 5F region were aligned and need Ops check off. The cameras for SLM's 2F09, 5F03 and 4f00 have been replaced and aligned. Repaired the valve assembly hardware for SLM5F03.  Assembly has begun for the new SLM at 4F09. Completed fabrication of another AMS chassis; testing of the chassis will begin early next week. Progress continues on the development of the multi pass BPMs.  The electronic designs have been completed and both switch sets are in the process of being installed.  The timing for the system will be supplied from a timing card that needs to be commissioned and tested. 
    Cables were pulled for the Optics CAN for OTS lite.  These cables provide LVDT signals and stepper motor control.  All cables were terminated and verified.  Planning for multipass BPM's capabilities continues.  Switch characterization is in progress and plans for implementation are being developed.  Future design for next generation BPM's is also being considered. The first draft of documentation for User Lab 1 LPSS PLC has been completed.  Installation of the thermal interlocks and modification of the interlock chassis was completed and tested for the suppression chicane magnet string. 2 RTD chassis' for the wiggler interlocks were programmed and installed. The box supply was brought online successfully.  All trim rack power supplies were brought on line this week and re-commissioned. The new Sextupoles (MSD2F07, 2F09) wee powered up and tested. We will still have to provide air cooling for these. The quad and corrector magnet power leads in the 4F region have been re-installed and re- commissioning is in progress.
WBS 9 (Beam Transport):

Installing the EM Wiggler and Decompression Chicane to Reduce Downstream THz 

•  Installation is complete except for punch list items. 

Sextupole (SF) 

•  Procurement package is out to vendors for bid. 

Replacement Chicane Dipoles GW) 

•  Procurement package is out to vendors for bid. 

UV Line 

• We are matching the existing surplus IR Demo girders to the available positions in the new line and are creating the modification drawing to convert the existing surplus IR Demo stands to the stands required for this line.
WBS 11 (Optics): 

2.8 and 1.06 micron mirrors: 
The 2.8 micron HR was installed and aligned.  The OC mirror will be installed next Monday.  Substrates for the 1.06 micron optics will be shipped to the vendor today.  We've received our 2.8 micron mirrors.  They passed all specifications.  A HR mirror was mounted onto its deformer.  An OC mirror was mounted into a new frame (made of niobium instead of molybdenum) with a new cooling line geometry.  To improve wetting of the low temperature braze between the mirror and frame, we are magnetron sputtering gold onto the frame and the edge of the optic.  Thus, 2.8 micron optics will be ready for installation.  As for 1.06 micron optics, an award was made with a vendor of known, low-loss coatings.  

 

Other activities:
We made great progress on this last week devoted primarily to installation activities.  As mentioned above, the new HR was mounted, and preparations were made to install the absorber plates.  The cryopump installation was completed.  Both ultraviewers were aligned to the wiggler viewers.  We also aligned the relay optics for the wiggler viewers to make this possible.  The optical transport system (OTS Lite) is nearly complete up to the Optical Control Room (OCR), and will be completed early next week.  Flat mirrors were mounted to their backplane coolers and installed in the collimator, and the three mirror cans in the vault.  All spool pieces and bellows were leak-checked this week, they will be installed (along with their stiffener rods) so we can pull a vacuum on the OTS.  A pumpout port and viewport are being installed in User Lab 1 for a laser personnel safety system (LPSS) check.  Diagnostics for the OCR are undergoing checkout - we thank the research group from the College of William and Mary for letting us use the output of their OPA system to calibrate our PbSe array for the spectrometer.  Work on a new, fast scanning (~ 10 Hz) autocorrelator progresses, as does the design for the optical beam positioning system.

Terahertz Project:

The remote control mount for M2 was completed ready for testing, and the figure of M2 was measured and found to meet specifications.  A mount for a camera to look at the first shutter/viewer was installed.  The manually adjustable mounts for mirrors M3 and M4 were cleaned and assembled after some modifications to the motion mechanism. Modifications to the beam conditioning vacuum box were completed. 

